Author's response to reviews

Title: What stops children with a chronic illness accessing health care: a mixed methods study in children with Chronic Fatigue Syndrome/Myalgic Encephalomyelitis (CFS/ME)

Authors:

Carly M Webb (m0702395@sgul.ac.uk)
Simon M Collin (simon.collin@bristol.ac.uk)
Toity Deave (toity.deave@uwe.ac.uk)
Andrew Haig-Ferguson (andrew.haig-ferguson@nhs.net)
Amy Spatz (aspatz@sgul.ac.uk)
Esther Crawley (esther.crawley@bristol.ac.uk)

Version: 5 Date: 11 November 2011

Author's response to reviews:

Author list: corrected for consistency between manuscript and submission system
Tables: moved to a tables section following the references
Tables: visible vertical lines removed from table
Typography: checked throughout - some corrections made